Author's response to reviews

Title: Injury prevention briefings: a strategy to aid implementation of community public health interventions among at risk populations? Qualitative study nested within a multi-centre randomised controlled trial

Authors:

Kate Beckett (Kate2.Beckett@uwe.ac.uk)
Trudy Goodenough (Trudy.Goodenough@uwe.ac.uk)
Toity Deave (Toity.Deave@uwe.ac.uk)
Sally Jaeckle (Sally.Jaeckle@bristol.gov.uk)
Lisa McDaid (Lisa.Mcdaid@nnuh.nhs.uk)
Penny Benford (Penny.Benford@nottingham.ac.uk)
Mike Hayes (Mike.Hayes@capt.org.uk)
Elizabeth Towner (Elizabeth.Towner@uwe.ac.uk)
Denise Kendrick (Denise.Kendrick@Nottingham.ac.uk)

Version: 5 Date: 23 October 2014

Author's response to reviews:

Please see detailed response to reviewers comments appended as a table to the covering letter